



Lessons Learned From Translational Research in Neuromuscular Diseases: Impact on Study Design, Outcome Measures and Managing Expectation

Georgia Stimpson¹, Mary Chesshyre^{1,2}, Giovanni Baranello^{1,2} and Francesco Muntoni^{1,2}*

¹Developmental Neuroscience Research and Training Department, Dubowitz Neuromuscular Centre, Faculty of Population Health Sciences, UCL Great Ormond Street Institute of Child Health, University College London, London, United Kingdom, ²NIHR Great Ormond Street Hospital Biomedical Research Centre, UCL Great Ormond Street Institute of Child Health, University College London, London, United Kingdom

OPEN ACCESS

Edited by:

Leslie Nelson, University of Texas Southwestern Medical Center, United States

Reviewed by:

Rachel Yaffa Zisk Rony, Hadassah Medical Center, Israel Lindsay Alfano, Nationwide Children's Hospital, United States

*Correspondence:

Francesco Muntoni f.muntoni@ucl.ac.uk

Specialty section:

This article was submitted to ELSI in Science and Genetics, a section of the journal Frontiers in Genetics

Received: 17 August 2021 Accepted: 09 November 2021 Published: 07 December 2021

Citation:

Stimpson G, Chesshyre M, Baranello G and Muntoni F (2021) Lessons Learned From Translational Research in Neuromuscular Diseases: Impact on Study Design, Outcome Measures and Managing Expectation. Front. Genet. 12:759994. doi: 10.3389/fgene.2021.759994 Spinal Muscular Atrophy (SMA) and Duchenne Muscular Dystrophy (DMD), two of the most common, child onset, rare neuromuscular disorders, present a case study for the translation of preclinical research into clinical work. Over the past decade, well-designed clinical trials and innovative methods have led to the approval of several novel therapies for SMA and DMD, with many more in the pipeline. This review discusses several features that must be considered during trial design for neuromuscular diseases, as well as other rare diseases, to maximise the possibility of trial success using historic examples. These features include well-defined inclusion criteria, matching criteria, alternatives to placebo-controlled trials and the selection of trial endpoints. These features will be particularly important in the coming years as the investigation into innovative therapy approaches for neuromuscular diseases continues.

Keywords: spinal muscular atrophy, duchenne and becker muscular dystrophy, translational research, trial design, inclusion criteria, matching criteria

INTRODUCTION

Translational research in neuromuscular diseases has evolved dramatically over the past few years, with dozens of clinical trials in the pipeline for several conditions, and a few approved novel treatments that are now available to patients in clinic. This rapid expansion of translational research and therapeutics has not been devoid of pitfalls and disappointments; however, these missteps have led to customized trial outcome measures and improved clinical trial design for neuromuscular diseases. The lessons learned from clinical trials, and the wider implementation of approved treatments in the real world, are ultimately moving the neuromuscular field towards individualised treatments based on patient characteristics. Two neuromuscular diseases, Spinal

Abbreviations: 6MWD, 6 minute Walking Distance; AAV, Adeno Associated Virus; ASO, Antisense Oligonucleotide; CHMP, Committee for Medicinal Products for Human Use; DMD, Duchenne Muscular Dystrophy; EMA, European Medicines Agency; FDA, US Food and Drug Administration; GC, Glucocorticoid; HFMSE, Hammersmith Functional Motor Scale Expanded; MCID, Minimal Clinically Important Difference; MDC, Minimal Detectable Change; NSAA, North Star Ambulatory Assessment; PODCI, Paediatric Outcomes Data Collection Instrument; PRO, Patient Reported Outcome; RCT, Randomised Control Trial; RULM, Revised Upper Limb Measure; RWE, Real World Evidence; SMA, Spinal Muscular Atrophy; SMN, Spinal Motor Neurone; SoC, Standard of Care.

Muscular Atrophy (SMA) and Duchenne Muscular Dystrophy (DMD) present case studies of the massive work done over the past two decades that has translated preclinical into clinical research. This work has included developing new diseasespecific outcome measures, well-designed clinical trials, and identifying the optimal patient population to be investigated in the clinical trials.

Background

SMA is an autosomal recessive condition with an incidence of 1 in 6,000 to 20,000 live births (Jones et al., 2015). SMA is caused by low levels of survival motor neuron (SMN) protein, resulting from mutations in the SMN1 gene. The paralog SMN2 gene is only able to make markedly reduced functional SMN protein, approximately 10% of the levels produced by SMN1, due to skipping of exon 7, leading to an ineffective protein. In SMA, muscle weakness and atrophy predominantly affect the proximal lower limbs, followed by a progressive decline of strength in the upper limbs and axial muscles (Wadman et al., 2018). SMA types are defined by age at symptom onset and the maximum motor milestone achieved. Those with SMA I have symptom onset before 6 months and never sit independently, in fact very few achieve motor milestones such as head control or rolling (Bruggen, 2016; de Sanctis et al., 2016). Those with SMA II have symptom onset between 7 and 18 months, and can sit independently but never walk unaided, while those with SMA III have symptom onset after 18 months and can walk independently, although they can lose this ability with time. SMA IV has onset during adulthood (Mercuri et al., 2012). In general, individuals with SMA II/III may experience an improvement in motor function in the first few years of life, until the ages of five and six to seven respectively (Wadman et al., 2018; Coratt et al., 2020; Coratti et al., 2020); this is usually followed by a fairly stable period and then a non-linear decline with time (Mercuri et al., 2016) which can be difficult to capture using some of the motor function assessment methods (Kaufmann et al., 2011; LoMauro et al., 2016).

DMD is an X-linked disease affecting approximately 1 in 5,000 male live births (Ryder et al., 2017). DMD is a progressive disorder characterised by muscle wasting and weakness. The eight reading frame rule states that DMD mutations that disrupt the reading frame and lead to loss of dystrophin protein are associated with DMD, whilst in-frame DMD mutations that lead to a reduction in the amount or size of dystrophin protein are associated with the milder Becker muscular dystrophy (Bladen et al., 2015). The reading frame rule has been reported to hold true for DMD in 93% in the Treat NMD global database, 91% in the Leiden DMD database and 96% in the French UMD DMD database (Aartsma-Rus et al., 2006; Tuffery-Giraud et al., 2009; Bladen et al., 2015). However, in the Leiden DMD database, Aartsma-Rus et al. (2006) observed that when DMD mutations were confirmed at the RNA level, rather than just the DNA level, over 99.5% fit with the reading-frame rule (Aartsma-Rus et al., 2006). Notable exceptions include large in-frame deletions and duplications, or even small in-frame deletions or other mutations affecting critical domains of dystrophin, such as the dystroglycan binding domain.

Boys with DMD are typically diagnosed between the age of 3-5 years (Bushby et al., 1999; Mohamed et al., 2000; Ciafaloni et al., 2009; van Ruiten et al., 2014) and experience a continuous decline in motor function after approximately age 7 (Mazzone et al., 2011; Ricotti et al., 2016). In glucocorticoid (GC) naive boys, ambulation is typically lost before the age of 12. The chronic use of GCs (Prednisone/Prednisolone or Deflazacort) has been shown to prolong ambulation up to the age of 12-15 years (Ricotti et al., 2013; Sussman et al., 2020; Zhang et al., 2021). In addition to progressive muscle weakness in the lower limbs leading to loss of ambulation, DMD boys also show progressive weakness of the upper limbs, respiratory and cardiac muscles, which leads to cardiomyopathy and respiratory insufficiency in later life (Spurney, 2011; Ricotti et al., 2019; Bello et al., 2020). Patients with DMD have a reduced life expectancy, with a mean age of survival in the late 20s (Eagle et al., 2002; Eagle et al., 2007; Passamano et al., 2012). Both Ricotti et al. (2016) and Darmahkasih et al. (2020) have noted high rates of comorbidities of neurodevelopmental, behavioural and emotional symptoms, including intellectual disabilities, anxiety and inattention.

The treatment landscapes for SMA and DMD have evolved dramatically in recent years. Three disease-modifying treatments have been authorised by the US Food and Drug Administration (FDA) and European Medicines Agency (EMA) for SMA, Onasemnogene abeparvovec (ZolgensmaTM), Risdiplam (EvrysdiTM) and Nusinersen (SpinrazaTM). Onasemnogene abeparvovec is a one-time genetic replacement adeno associated virus (AAV) therapy that contains a functional *SMN1* gene. Nusinersen and Risdiplam both act on the splicing of the *SMN2* gene and induce the retention of exon 7 in the transcript, hence allowing the increase in the production of functional SMN. Risdiplam is a daily oral medication, whereas Nusinersen is administered intrathecally via lumbar puncture, three times in a year after the loading phase in the few months that requires more frequent administration.

While GCs remain the mainstay of pharmacological treatment for DMD, several DMD mutation-specific treatments have been developed. One of these, Ataluren, enables ribosomal readthrough of premature stop codons present in 10-15% of patients with nonsense mutations (Bushby et al., 2014). It has been approved by EMA for use in the EU (Committee for Medicinal Products for Human Use (CHMP), 2018a). In separate efforts, several groups have developed exon skipping therapies, which aim to restore the DMD reading frame isoforms in patients with out-of-frame deletions and to allow the expression of partially functional dystrophin (Niks and Aartsma-Rus, 2017). Several such antisense oligonucleotide (ASO) drugs, which require weekly intravenous administration, have been developed in the last decade, targeting exon 51 [Eteplirsen (Mendell et al., 2013; Mendell et al., 2016)], exon 53 [Golodirsen (Frank et al., 2020) and Viltolarsen (Clemens et al., 2020)] and exon 45 [Casimersen (Wagner et al., 2021)]. 14, 10, and 9% of all DMD boys are amenable to skipping of exons 51, 53, and 45, respectively (Bladen et al., 2015). All four of these drugs are approved in the US (Shirley, 2021; Heo, 2020; Syed, 2016; U.S. Food and Drug Administration, 2020) and Viltolarsen also in Japan (Dhillon, 2020), but they have not been approved yet in Europe. This is due to differences in the emphasis given to surrogate outcome measures (such as dystrophin restoration in muscle) between the different regulatory agencies.

IMPACT ON STUDY DESIGN

Inclusion Criteria and Participants Selection

Despite the exponential increase of clinical trials in SMA and DMD, both conditions present significant statistical challenges, due to the systemic nature of the disease and the heterogeneity in clinical manifestations. For this reason, significant work has been done to optimise clinical trial design, and several lessons can be learnt from these previous experiences. When designing clinical trials in neuromuscular diseases (and other rare diseases), there is a trade-off between enrolling the subpopulations who are most likely to benefit from the treatment (the "optimal" population) and having less strict inclusion criteria to enrol a representative sample of the overall patient population. This is important to allow a broad label for the medicinal product so that not only those patients within the narrow margin of the trial inclusion criteria could benefit from them. Historically, the field has also faced several trials which were unsuccessful due to insufficient treatment effects, such as Phenylbutyrate (Mercuri et al., 2007), Valproic Acid (Swoboda et al., 2010), Raxone (Servais et al., 2020) and more recently the anti-myostatin trials (ClinicalTrials.gov: NCT03039686 and NCT02310763). For an overview of key features for all of the trials referenced in this review refer to Table 1.

This has highlighted the need to specify inclusion/exclusion criteria that will lead to the enrolment of a patient population with better defined natural history data or a higher likelihood of response to treatment. A case study for this was the experience in the Nusinersen sham-controlled trials for SMA. In the CHERISH study in patients with SMA II (Mercuri et al., 2018), strict inclusion/exclusion criteria were applied, including a relatively high functional motor score on the Hammersmith Functional Motor Scale Expanded (HFMSE) scale, between 10 and 54, a maximum age at enrolment of 12 years and the absence of contractures and severe scoliosis. These well-defined criteria were crucial to the success of the study, which ultimately led to the approval of Nusinersen.

The issue of identifying well-defined inclusion criteria, which take into account functional variability and different predicted long-term trajectories, has become obvious in recent years for DMD. Almost invariably all recent trials in DMD only include patients receiving standard of care (SoC) treatments, such as those being on a stable dose and regime of GCs (see ClinicalTrials.gov: NCT02255552, NCT01826487 and NCT04281485 for example). This can lead to potential bias in patient selection, particularly in younger boys. In DMD, disease severity has been correlated with GC starting age, with patients with more severe phenotypes more likely to start GCs at a younger age (Ricotti et al., 2016). Similarly, Muntoni, et al. (2019) used latent class trajectory analysis to cluster DMD

boys into four groups based on the North Star Ambulatory Assessment (NSAA) score. They found that those in the two clusters with fastest disease progression (class 1 and class 2) had an earlier mean age of starting GCs than those in the two clusters with slower disease progression. This suggests that by preferentially offering GCs to young DMD boys with early clinical presentation there is a risk of bias towards selecting those on a more severe trajectory. Removing these criteria means a more representative sample of DMD patients, although it leads to trajectory heterogeneity which represents an added difficulty for disease progression modelling.

To avoid these complexities, a broader population of patients, with a well-defined subpopulation with stricter clinical and functional inclusion/exclusion criteria has been shown in the literature as a viable approach. In the ACT DMD trial (McDonald et al., 2017) for Ataluren, a sub-population of interest, those boys whose 6 Minute Walking Distance (6 MWD) was between 300 and 400 m, was pre-specified, but there was also a broader enrolment. The argument for this specification was that patients who have better functional abilities (>400 m 6 MWD) remain stable over the 48 weeks of the study. Therefore, as Ataluren aims to induce stability, this population were not optimal for demonstrating treatment effect. On the other hand, it was argued that the weakest patients (6 MWD <300 m) had a high rate of interpatient variability, often having started the very rapid phase of disease progression leading to loss ambulation within a short period of time. This variability and different stages of the disease can cloud the ability to demonstrate a treatment effect. By pre-specifying a subpopulation of interest, but also having broader inclusion criteria, the researchers maximised the chance of getting clear estimates of treatment effect, whilst also beginning to understand the variation in treatment effect. This study showed that in the whole trial population, the treated group lost on average 47.7 m over the 48 weeks follow up, whilst the placebo group lost on average 60.7 m. There was no significant difference between these two groups (p = 0.213). However, in the prespecified group with 6 MWD between 300 and 400 m, the treated group lost on average 27.0 m over the 48 weeks follow up, whilst the placebo group lost on average 69.9 m. Here the difference was significant (p = 0.007). A similar, post-hoc subgroup analysis was also done in the phase 3 trial of the DMD gene exon 51 skipping ASO Drisapersen (Goemans et al., 2018). Here, the outcome measure of interest was also the 6 MWD but there was significantly more variability in the data than was expected, leading to an overestimation of the trial power and a non-significant treatment effect. Consequently, a secondary analysis, on only those patients with a 6 MWD between 300 and 400 m who could rise from the floor, was completed. This yielded a significant treatment effect of 35.4 m (p = 0.039) for a 48 weeks follow up. However, this restrictive criteria approach can underestimate the treatment effect heterogeneity, due to bias sampled into the trial population. This can then cause issues for clinicians in managing patient and carer expectations of new treatments.

Matching between the treated and placebo arms is critical to allow the detection of treatment effects. Several factors can impact the treatment effect and interpretation of clinical trial results, including age, disease duration and functional status at baseline. A primary example of this is the recently announced top-line results of the Sarepta micro-dystrophin gene-therapy trial (Sarepta Therapeutics, 2021). No significant difference (p = 0.37) was found between the change in NSAA score for the treated and placebo arms. The investigators suggested that this may be due to a significant difference in the motor function scores at baseline between the patients aged 6 to 7, where treated patients had significantly worse scores not only on the NSAA scores, but raise time to stand from supine, 10 m walk and 100 m run at baseline compared to those on placebo (p = 0.005). This may have been the reason that the treated arm did not show a drug effect compared to the placebo group, as the patients who were in the treated group were more severe at baseline.

Factors That Contribute to Disease Heterogeneity

There is significant heterogeneity of disease progression in both SMA and DMD, as well as in treatment approaches, which warrant significant consideration when designing clinical trials. It is worth noting here that the degree of variability in the patient population that is tolerable in a trial is directly related to the size of the treatment effect and the extent to which this is captured by the outcome measure.

The impact of genetic variability on this variation is not fully understood as of yet. In SMA, the number of copies of *SMN2* has been reported to correlate with disease severity but not with response to treatment (Baranello et al., 2021). However, clinical severity and treatment response vary significantly between patients with SMA, as several factors, including the interval between disease onset and initiation of therapy, and intragenic and structural variation in the *SMN* locus have been reported to possibly explain at least partly the clinical variability in SMA (Wadman et al., 2020).

In recent years, the genetic background and the differential expression of dystrophin isoforms have been proposed as key factors in the design of clinical trials (Desguerre et al., 2009). Several studies have shown that more distal mutations in the DMD gene are correlated with worse respiratory and cardiac outcomes (Yamamoto et al., 2018; Bello et al., 2020). The recent long term natural history studies in DMD patients with different genetic backgrounds have contributed to understanding motor function trajectories for DMD patients with different mutations amenable to exon skipping; those amenable to skip exons 53 and 51 display lower motor function scores at baseline and more negative changes than the other subgroups (Brogna et al., 2018; Bello et al., 2020). When it comes to recruiting populations for randomised control trials (RCTs), considerable effort should be made to ensure that the treated and placebo populations are matched in terms of genetics, along with other more standard criteria such as age and baseline motor function.

Genotype-phenotype correlation presents an additional level of difficulty in getting well matched trial populations when the trial population is further restricted. Exon skipping drugs in DMD provide an example of this, as the drugs are only appropriate for a subset of the DMD population: for example, those eligible for exon 51 skipping drugs make up ~14% of the DMD population. It has been shown that amenability to different exon-skipping drugs correlates with disease trajectory (Brogna et al., 2018; Bello et al., 2020), and therefore it is crucial that treated and placebo patients are matched according to *DMD* genotype.

Matching on SoC can also present significant challenges in trial design. For example, in DMD GCs form much of the SoC treatments, and several studies have shown that GC use delays loss of ambulation (Bushby et al., 2004). There are, however, large amounts of variety in GC regimes, including variation in type, frequency, and dose (Griggs et al., 2013). Clinicians often prescribe specific regimes based on patient characteristics, but the true effect of each regime has not been fully understood. These differences in GC treatment add an additional level of complexity to matching in trials, especially as it is known that different GC regimes may have more significant effects on motor outcomes, such as loss of ambulation (Ricotti et al., 2013).

Beyond the issue of matching cohorts as presented above, traditional RCTs for DMD and SMA present other difficulties. The presence of a placebo arm in an RCT can be unattractive to patients and carers if the risk (typically 33% or 50%) of being randomised to the placebo arm is perceived as too high. In SMA in particular, where other disease-modifying therapies have already been approved, ethical considerations are necessary if placebo arms are included in trials. This is even truer for patients with SMA I where earlier age of treatment relative to symptom onset has been shown to result in more favourable outcomes (Finkel et al., 2017). Due to the increasing availability of disease-modifying treatments to SMA patients, future trials will need to adapt to this evolving landscape and to consider patients treated in the real world as comparators.

Novel Trial Design With Real-World Data

The aforementioned challenges in recruiting both SMA and DMD patients into RCTs with placebo arms and sufficient duration of follow-up to detect treatment effects has led to an interest in alternative trial designs within the neuromuscular community. These alternative trial designs do not require a unique, well-matched placebo arm, but instead, aim to draw on previous data on untreated (or SoC treated) patients who can be "selected" into an augmented control arm by applying the same inclusion criteria. This is referred to as real-world evidence (RWE) by the FDA and guidance for trial design using RWE has been published recently (U.S. Department of Health and Human Services and Food and Drug Administration, 2019). There are two types of untreated populations that can serve as comparators, placebo arms from other trials and natural history studies. Both populations have their advantages and disadvantages. Historic placebo arms from completed trials are often formatted in a way that makes them easy to use as augmented controls. This is because the outcome measures for trials in a specific disease are often the same, and these outcomes are assessed at regular followups with a very low proportion of missing data. However, these historic placebo arms are often fairly small, and the duration of the trials is limited, so it may become necessary to use multiple different trial placebo arms. As inclusion criteria for trials tend to

be fairly restrictive, there can also be bias if the historic placebo arm (or arms) does not have sufficiently similar inclusion criteria. On the other hand, natural history data are often very large, and the trial inclusion criteria can be applied to datasets to create an augmented control arm. There are limitations with this approach, as there is an inherent bias in natural history data, as clinical decisions for patients are not randomised. Additionally, the natural history data itself can have high proportion of missing data and inaccuracies. Applying trial inclusion criteria to natural history data is not straightforward, particularly because inclusion into a new therapy trial is often a proxy measure for disease severity, with those included in trials often younger and with the most typical disease progression (Franklin and Schneeweiss, 2017). Goemans et al. (2020) has shown that change in 6 MWD, one of the most used DMD outcome measures, is consistent between six placebo arm groups and four natural history observational studies, suggesting that the choice of which set of data to use might not be so critical, and that the use of both natural history data and historic placebo arms to create augmented control arms might become more common in future trials. In keeping with current regulatory requirements, it is worth noting that if an augmented control arm is to be considered it must be specified prospectively in the trial design and should not be done retrospectively if the actual placebo arm is deemed insufficient (Committee for Medicinal Products for Human Use (CHMP), 2018b).

Lake et al. (2021) described a novel, practical method for incorporating placebo arms from other trials into the analysis of the Dystance 51 phase 2/3 trial of Suvodirsen (ClinicalTrials.gov NCT03907072), another ASO for the skipping of exon 51 of the DMD gene. Unfortunately, the Dystance 51 trial was stopped due to the lack of increased dystrophin expression in the open-label extension of the phase 1 Suvodirsen trial (Wave Life Sciences, 2019). This approach, using an adapted Bayesian trial design, would have been implemented for the analysis if the trial had been completed. Despite having a placebo arm itself, the design of this trial was such that the placebo arms from three historic trials ("historic placebos") could be used to augment those randomised to placebo in the Dystance 51 trial. This augmentation was done by Bayesian meta-analysis, whereby the information from the historic placebos was dynamically "borrowed" into the augmented placebo arm. The amount of borrowing depends on how well the true placebo arm mirrored the historic placebo arms in terms of the primary endpoint. For example, if the placebos in the Dystance 51 trial act similarly to the historic placebos then more information can be borrowed into the augmented placebo arm, but if the placebo arms look different then less is borrowed. This method results in fewer patients needing to be recruited into the placebo arm to achieve the effective sample size needed in the trial.

Clinically Relevant Endpoints

A key difficulty when designing trials for DMD and SMA is how to define an endpoint that is feasible, meaningful, and timely. In DMD for example, there are three outcomes of interest: biological, where a significant increase in dystrophin is produced in treated patients compared to untreated patients; clinical, where a significant change in clinical outcome measures are observed relative to the natural history or placebo cohorts; and patient reported, where a significant change is perceived as a clinically-relevant endpoint by patients and/or their family/ carers. The FDA and EMA both have accelerated approval pathways for drugs developed to treat serious or lifethreatening conditions. The FDA accepts the use of surrogate, biological trial endpoints that are considered reasonably likely to predict benefit to patients, with confirmation of clinical benefit for these drugs in the post-marketing setting as a condition of continued approval. By contrast, EMA considers that at present there are no suitable biomarkers to establish treatment efficacy and data on a clinical endpoint are required for approval.

Eteplirsen, an ASO skipping exon 51, is a crucial case study here, as it has been approved in the US but not in Europe (Committee for Medicinal Products for Human Use (CHMP), 2018b; US Food and Drug Administration, 2016). The PROMOVI Eteplirsen trial (ClinicalTrials.gov NCT0225552) found a significant difference in dystrophin production between the placebo and treated group but failed to find a significant difference in the change in motor function between the two groups. The question then arose, what is the threshold above which an increase in dystrophin levels will have a clinical effect and is it enough that a drug can be shown to work biologically but not clinically. This was one of the main reasons for the EMA CHMP rejection.

Although not included in DMD and SMA trials as the primary endpoint, patient reported outcomes (PROs) are becoming more common as secondary outcomes in recent clinical trials. For example, they are included in the Sarepta Micro-dystrophin for DMD trial (ClinicalTrials.gov NCT03769116), the Pfizer minidystrophin for DMD trial (ClinicalTrials.gov NCT04281485) and the ACT-DMD Ataluren trial (ClinicalTrials.gov NCT01826487). Although the most commonly used PRO measure in DMD is the Paediatric Outcomes Data Collection Instrument (PODCI) (Daltroy et al., 1998), there has been a recent drive to develop specialised PRO measures for specific disease stages for both DMD and SMA. For example, the DMD specific Quality of Life (QoL) measure, the DMD-QoL (Powell et al., 2021), and the DMD upper body specific PRO measure, the DMD Upper Limb PROM (Klingels et al., 2017), have been developed specifically for the DMD population. Additionally, both the FDA and EMA have published reports reiterating the need for patient reported outcomes for supporting claims in product labelling (Committee for Medicinal Products for Human Use (CHMP), 2005; FDA Center for Biologics Evaluation and Research, 2009).

A key issue with clinical outcome measures is ensuring that the observed change is meaningful, and we discuss here several different issues relating to that. First, we must be able to say that the observed change is real, not just variability in the measure. One part of this is knowing the minimal detectable change (MDC), for which novel methods have been developed. We must also be aware of the reliability of the score, and particularly between different centres and different assessors. The MDC is the smallest difference that indicates a real clinical change in disease. This value for 80% confidence that a change is indicative of a true change has been calculated by

TABLE 1 | Overview of key trials in DMD and SMA discussed in this review.

Trial Number (Name)	Trial drug	Biology	Trial stage	Trial population	Trial features	Primary endpoint	Maximum follow up	N	Complete (mm/yy)
DMD NCT03769116	SRP-9001	AAV mediated gene therapy with micro-dystrophin	2	 4–7 years, Stable GCs for 12 weeks prior 	Multicentre, randomised, placebo-controlled cross-over trial for efficacy	Change in Micro- Dystrophin protein expression (western blot) Change in NSAA	48 weeks	41	No
NCT04281485	PF-06939926	AAV mediated gene therapy with mini dystrophin	3	 4–7 years old, Ambulant, Stable GCs for 3 months prior 	Randomised (2:1), double-blind, placebo-controlled efficacy trial	NSAA change from baseline	52 weeks	~99	No
NCT02310763	Domagrozumab (PF-06252616)	Anti-myostatin adnectin	2	• $6-16$ years, • $4SC \le 6.4$ s, ≥ 1.32 s, • GC use for ≥ 6 months prior, stable GCs for 3 months prior	Randomized, double-blind, placebo-controlled, multi-dose, efficacy and safety study	Adverse events Change in 4SC	49 weeks	121	Yes (08/18)
NCT03039686	RG6206 (RO7239361)		2/3	 6–11 years, Ambulant (no assistance), NSAA≥ 15 at baseline 4SC ≤8 s, GC use 	Randomized, double-blind, placebo-controlled, multi-dose, efficacy and safety study	Change in NSAA score	48 weeks	166	Yes (12/20)
NCT02255552 (PROMOVI)	Eteplirsen	PMO for exon 51 skipping	3	 7–16 years, Stable GCs for 24 weeks prior, 6 MWD ≥300 m (main analysis 300–450 m), Stable pulmonary and cardiac function 	Open-label, multi- centre, study with concurrent control arm (not treated, not amenable) for efficacy and safety	Change in 6 MWD	96 weeks	109	Yes (01/21)
NCT03907072	WVE-210201 (Suvodirsen)		2/3	 5–12 years, Ambulant, Stable pulmonary and cardiac function, Amenable to exon 51 skipping, GC use for >6 months prior, stable GCs for 3 months prior 	Randomized, double-blind, placebo-controlled, multi-dose, efficacy and safety study with open-label extension	Change in Dystrophin Level and NSAA score	48 weeks	6	Yes (05/21)
NCT01826487 (ACT-DMD)	Ataluren	Small molecule that restores dystrophin synthesis by allowing ribosomes to read through premature stop codons	3	 7–16 years, Nonsense mutation DMD, GC use for >6 months prior, stable GCs for 3 months prior, 6 MWD ≥150 m, 6 MWD ≤80% predicted for age 	Efficacy and Safety Study placebo controlled	6 MWD	48 weeks	230	Yes (08/20)
SMA NCT02292537 (CHERISH)	Nusinersen (ISIS 396443)	SMN2-directed RNA splicing modifier	3	 2–12 years, Onset ≥6 months old, HFMSE ≥ 10 and ≤ 54 at baseline, Independent sitter, Never ambulant 	Randomised, double-blind, sham- procedure controlled study to assess efficacy and safety	Change in HFMSE score	15 months	126	Yes (02/21)

Muntoni et al. (2021) for the NSAA (2.78), 6 MWD (36.3), 4 stair climb (4SC) time (1.31) and 4SC velocity (0.35).

We must also take into account the inherent variability in outcome measures. This can take the form of test-retest reliability, such as in biological outcomes, where natural variation in biomarkers means that they can only be assessed with error. Additionally, inter-rater reliability can be of concern, such as in motor function assessments where there can be a discrepancy between assessors. Additionally, variation in methodology for the extraction and analysis of biomarkers can also lead to inter-rater reliability issues. Finally, the use of clinical and patient reported outcomes in populations for which the measure has not been validated can lead to significant error. All three of these factors can lead to increased noise when assessing the treatment effect.

The minimal clinically important difference (MCID) must also be considered when selecting the outcome of a clinical trial. Ideally, any reported outcome of a trial would be greater than the MCID to indicate a meaningful effect on patient quality of life. To assess true MCID patients and carers are often contacted, and their opinions assessed via questionnaires. Alternative methods to calculate the MCID include anchoring the score of interest to other scores which already have a validated MCID or calculating the MDC as a proxy for the MCID using distribution-based approaches. For example, Stolte et al., 2020 reported the MDC (as a proxy for the MCID) for the revised upper limb module (RULM) and the HFMSE between 2.9 and 6.4 and between 4.3 and 10.6 respectively. The linear NSAA is a transformed version of the NSAA ranging from 0-100, where the change in scores are standardised across the scale (Mayhew et al., 2013). The MDC (as a proxy for the MCID) for the linear NSAA in DMD was found to be 8.8 and 6.9 for boys on daily and intermittent prednisolone respectively (Mayhew et al., 2013), whilst for the 6 MWD in DMD it was found to be 28.5/31.7 (Mcdonald et al., 2013).

CONCLUSION

In conclusion, translational research in neuromuscular diseases has had a pioneering role over the past decade in developing high quality, well-designed clinical trials that have led to the approval of several novel therapies. Much has been learnt on how to select patients to be recruited based on foreseen trajectories and possible responses to treatment. With the rapidly evolving technology and translation of preclinical science to patients, it will be crucial to

REFERENCES

- Aartsma-Rus, A., van Deutekom, J. C. T., Fokkema, I. F., van Ommen, G.-J. B., and den Dunnen, J. T. (2006). Entries in the Leiden Duchenne Muscular Dystrophy Mutation Database: An Overview of Mutation Types and Paradoxical Cases that Confirm the reading-frame Rule. *Muscle Nerve* 34 (2), 135–144. doi:10.1002/mus.20586
- Baranello, G., Gorni, K., Daigl, M., Kotzeva, A., Evans, R., Hawkins, N., et al. (2021). Prognostic Factors and Treatment-Effect Modifiers in Spinal Muscular Atrophy. *Clin. Pharmacol. Ther.* doi:10.1002/cpt.2247

have both robust clinical trial design and adequate treatment effect to ensure the success of future clinical trials. With many failed trials, the increasing availability of treatment to patients in the real world, and the long duration that many studies require to detect a clinical response, alternative study designs are receiving increasing consideration by industries and regulatory authorities. These include comparison with natural history cohorts and placebo arms from other trials that may become particularly relevant for rare diseases in the near future. The proper design of a clinical trial is crucial to its success in leading to the approval of the medicine products under investigation; however, the implementation of approved treatments in the real world, where there are no strict inclusion/exclusion criteria and patients' clinical heterogeneity may be much wider, requires careful discussion and counselling with patients and caregivers to properly manage expectations. Post-marketing disease registries and databases will be essential in the coming years to collect real-world data. This in turn will help clinicians, carers, and patients alike to understand the short and long-term impact of these new neuromuscular therapies, as it has been done historically with renal and cancer registries.

AUTHOR CONTRIBUTIONS

GS and MC wrote the manuscript with comments, guidance in review direction and critical review provided by GB and FM.

FUNDING

GB and GS are supported by a grant from the Great Ormond Street Hospital Children's Charity and Muscular Dystrophy UK. FM and MC are supported by the NIHR Great Ormond Street Biomedical Research Centre. The views expressed are those of the author(s) and are not necessarily those of the NHS, the NIHR or the Department of Health.

ACKNOWLEDGMENTS

We are grateful to the DMD patients and their families and the SMA patients and their families. We are also grateful to Muscular Dystrophy UK, the North Star clinical network, the SMA Reach network and the iSMAC network.

Bello, L., D'Angelo, G., Villa, M., Fusto, A., Vianello, S., Merlo, B., et al. (2020). Genetic Modifiers of Respiratory Function in Duchenne Muscular Dystrophy. *Ann. Clin. Transl Neurol.* 7 (5), 786–798. doi:10.1002/acn3.51046

- Bladen, C. L., Salgado, D., Monges, S., Foncuberta, M. E., Kekou, K., Kosma, K., et al. (2015). The TREAT-NMD DMD Global Database: Analysis of More Than 7,000 Duchenne Muscular Dystrophy Mutations. *Hum. Mutat.* 36 (4), 395–402. doi:10.1002/humu.22758
- Brogna, C., Coratti, G., Pane, M., Ricotti, V., Messina, S., D'Amico, A., et al. (2018). Long-term Natural History Data in Duchenne Muscular Dystrophy Ambulant Patients with Mutations Amenable to Skip Exons 44, 45, 51 and 53. *PLoS ONE* 14 (6), e0218683–9. doi:10.1371/journal.pone.0218683

- Bruggen, H. W. (2016). Mandibular Function in Neuromuscular disordersPhD Thesis. Nijmegen, Netherlands: Radboud University.
- Bushby, K., Finkel, R., Wong, B., Barohn, R., Campbell, C., Comi, G. P., et al. (2014). Ataluren Treatment of Patients with Nonsense Mutation Dystrophinopathy. *Muscle Nerve* 50 (4), 477–487. doi:10.1002/mus.24332
- Bushby, K., Hill, A., and Steele, J. (1999). Failure of Early Diagnosis in Symptomatic Duchenne Muscular Dystrophy. *The Lancet* 353 (9152), 557–558. doi:10.1016/ S0140-6736(98)05279-9
- Bushby, K., Muntoni, F., Urtizberea, A., Hughes, R., and Griggs, R. (2004). Report on the 124th ENMC International Workshop. Treatment of Duchenne Muscular Dystrophy; Defining the Gold Standards of Management in the Use of Corticosteroids 2-4 April 2004, Naarden, The Netherlands. *Neuromuscul. Disord.* 14, 526–534. doi:10.1016/j.nmd.2004.05.006
- Ciafaloni, E., Fox, D. J., Pandya, S., Westfield, C. P., Puzhankara, S., Romitti, P. A., et al. (2009). Delayed Diagnosis in Duchenne Muscular Dystrophy: Data from the Muscular Dystrophy Surveillance, Tracking, and Research Network (MD STARnet). J. Pediatr. 155 (3), 380–385. doi:10.1016/j.jpeds.2009.02.007
- Clemens, P. R., Rao, V. K., Connolly, A. M., Harper, A. D., Mah, J. K., Smith, E. C., et al. (2020). Safety, Tolerability, and Efficacy of Viltolarsen in Boys with Duchenne Muscular Dystrophy Amenable to Exon 53 Skipping. *JAMA Neurol.* 77 (8), 982–991. doi:10.1001/jamaneurol.2020.1264
- Committee for Medicinal Products for Human Use (Chmp) (2018a). Assessment Report - Translarna. Available at: https://www.ema.europa.eu/en/documents/ variation-report/translarna-h-c-2720-p46-0025-epar-assessment-report_en. pdf (Accessed July 12, 2021).
- Committee for Medicinal Products for Human Use (Chmp) (2005). Reflection Paper on the Regulatory Guidance for the Use of Health-Related Quality of Life (HRQL) Measures in the Evaluation of Medicinal Products. Available at: https://www.ema. europa.eu/en/documents/scientific-guideline/reflection-paper-regulatory-guidanceuse-healthrelated-quality-life-hrql-measures-evaluation_en.pdf (Accessed October 11, 2021).
- Committee for Medicinal Products for Human Use (Chmp) (2018b). Scientific Conclusions and Grounds for Refusal Presented by the European Medicines Agency - Exondys. Available at: https://www.ema.europa.eu/en/documents/ scientific-conclusion/exondys-epar-scientific-conclusions-grounds-refusal_en. pdf (Accessed July 12, 2021).
- Coratti, G., Lucibello, S., Pera, M. C., Duong, T., Muni Lofra, R., Civitello, M., et al. (2020). Gain and Loss of Abilities in Type II SMA: A 12-month Natural History Study. *Neuromuscul. Disord.* 30 (9), 765–771. doi:10.1016/j.nmd.2020.07.004
- Coratti, G., Messina, S., Lucibello, S., Pera, M. C., Montes, J., Pasternak, A., et al. (2020). Clinical Variability in Spinal Muscular Atrophy Type III. Ann. Neurol. 88 (6), 1109–1117. doi:10.1002/ana.25900
- Daltroy, L. H., Liang, M. H., Fossel, A. H., and Goldberg, M. J. (1998). The POSNA Pediatric Musculoskeletal Functional Health Questionnaire: Report on Reliability, Validity, and Sensitivity to Change. J. Pediatr. Orthopaedics 18 (5), 561–571. doi:10.1097/00004694-199809000-00001
- Darmahkasih, A. J., Rybalsky, I., Tian, C., Shellenbarger, K. C., Horn, P. S., Lambert, J. T., et al. (2020). Neurodevelopmental, Behavioral, and Emotional Symptoms Common in Duchenne Muscular Dystrophy. *Muscle Nerve* 61 (4), 466–474. doi:10.1002/mus.26803
- de Sanctis, R., Coratti, G., Pasternak, A., Montes, J., Pane, M., Mazzone, E. S., et al. (2016). Developmental Milestones in Type I Spinal Muscular Atrophy. *Neuromuscul. Disord.* 26 (11), 754–759. doi:10.1016/j.nmd.2016.10.002
- Desguerre, I., Christov, C., Mayer, M., Zeller, R., Becane, H.-M., Bastuji-Garin, S., et al. (2009). Clinical Heterogeneity of Duchenne Muscular Dystrophy (DMD): Definition of Sub-phenotypes and Predictive Criteria by Long-Term Follow-Up. *PLoS ONE* 4 (2), e4347. doi:10.1371/journal.pone.0004347
- Dhillon, S. (2020). Viltolarsen: First Approval. Drugs 80 (10), 1027-1031. doi:10.1007/s40265-020-01339-3
- Eagle, M., Baudouin, S. v., Chandler, C., Giddings, D. R., Bullock, R., and Bushby, K. (2002). Survival in Duchenne Muscular Dystrophy: Improvements in Life Expectancy since 1967 and the Impact of home Nocturnal Ventilation. *Neuromuscul. Disord.* 12 (10), 926–929. doi:10.1016/S0960-8966(02)00140-2
- Eagle, M., Bourke, J., Bullock, R., Gibson, M., Mehta, J., Giddings, D., et al. (2007). Managing Duchenne Muscular Dystrophy - the Additive Effect of Spinal Surgery and home Nocturnal Ventilation in Improving Survival. *Neuromuscul. Disord.* 17 (6), 470–475. doi:10.1016/j.nmd.2007.03.002

- Fda Center for Biologics Evaluation and Research (2009). Guidance for Industry Patient-Reported Outcome Measures: Use in Medical Product Development to Support Labeling Claims. Available at: https://www.fda.gov/media/77832/ download (Accessed October 11, 2021).
- Finkel, R. S., Mercuri, E., Darras, B. T., Connolly, A. M., Kuntz, N. L., Kirschner, J., et al. (2017). Nusinersen versus Sham Control in Infantile-Onset Spinal Muscular Atrophy. N. Engl. J. Med. 377 (18), 1723–1732. doi:10.1056/ nejmoa1702752
- Frank, D. E., Schnell, F. J., Akana, C., El-Husayni, S. H., Desjardins, C. A., Morgan, J., et al. (2020). Increased Dystrophin Production with Golodirsen in Patients with Duchenne Muscular Dystrophy. *Neurology* 94, e2270–e2282. doi:10.1212/ WNL.00000000009233
- Franklin, J. M., and Schneeweiss, S. (2017). When and How Can Real World Data Analyses Substitute for Randomized Controlled Trials? *Clin. Pharmacol. Ther.* 102 (6), 924–933. doi:10.1002/cpt.857
- Goemans, N., Mercuri, E., Belousova, E., Komaki, H., Dubrovsky, A., McDonald,
 C. M., et al. (2018). A Randomized Placebo-Controlled Phase 3 Trial of an
 Antisense Oligonucleotide, Drisapersen, in Duchenne Muscular
 dystrophyDuchenne Muscular Dystrophy. *Neuromuscul. Disord.* 28 (1),
 4–15. doi:10.1016/J.NMD.2017.10.004
- Goemans, N., Signorovitch, J., Sajeev, G., Yao, Z., Gordish-Dressman, H., McDonald, C. M., et al. (2020). Suitability of External Controls for Drug Evaluation in Duchenne Muscular Dystrophy. *Neurology* 95 (10), e1381–e1391. doi:10.1212/WNL.000000000010170
- Griggs, R. C., Herr, B. E., Reha, A., Elfring, G., Atkinson, L., Cwik, V., et al. (2013). Corticosteroids in Duchenne Muscular Dystrophy: Major Variations in Practice. *Muscle Nerve* 48 (1), 27–31. doi:10.1002/mus.23831
- Heo, Y.-A. (2020). Golodirsen: First Approval. Drugs 80, 329–333. doi:10.1007/ s40265-020-01267-2
- Jones, C., Oskoui, M., Zielinski, D., Vinikoor, L., and Farwell, W. (2015). PP09.1 -2352: Systematic Review of Incidence and Prevalence of Spinal Muscular Atrophy (SMA). *Eur. J. Paediatric Neurol.* 19, S64–S65. doi:10.1016/s1090-3798(15)30213-0
- Kaufmann, P., McDermott, M. P., Darras, B. T., Finkel, R., Kang, P., Oskoui, M., et al. (2011). Observational Study of Spinal Muscular Atrophy Type 2 and 3: Functional Outcomes over 1 Year. Arch. Neurol. 68 (6), 779–786. doi:10.1001/ archneurol.2010.373
- Klingels, K., Mayhew, A. G., Mazzone, E. S., Duong, T., Decostre, V., Werlauff, U., et al. (2017). Development of a Patient-Reported Outcome Measure for Upper Limb Function in Duchenne Muscular Dystrophy: DMD Upper Limb PROM. Dev. Med. Child. Neurol. 59 (2), 224–231. doi:10.1111/ dmcn.13277
- Lake, S. L., Quintana, M. A., Broglio, K., Panagoulias, J., Berry, S. M., and Panzara, M. A., 2021, Bayesian Adaptive Design for Clinical Trials in Duchenne Muscular Dystrophy. *Stat. Med.*, 40pp.4167–4184. doi:10.1002/sim.9021
- LoMauro, A., Aliverti, A., Mastella, C., Arnoldi, M. T., Banfi, P., and Baranello, G. (2016). Spontaneous Breathing Pattern as Respiratory Functional Outcome in Children with Spinal Muscular Atrophy (SMA). *PLoS ONE* 11 (11), e0165818–16. doi:10.1371/journal.pone.0165818
- Mayhew, A. G., Cano, S. J., Scott, E., Eagle, M., Bushby, K., Manzur, A., et al. (2013). Detecting Meaningful Change Using the North Star Ambulatory Assessment in Duchenne Muscular Dystrophy. *Dev. Med. Child. Neurol.* 55 (11), 1046–1052. doi:10.1111/dmcn.12220
- Mazzone, E., Vasco, G., Sormani, M. P., Torrente, Y., Berardinelli, A., Messina, S., et al. (2011). Functional Changes in Duchenne Muscular Dystrophy: A 12month Longitudinal Cohort Study. *Neurology* 77 (3), 250–256. doi:10.1212/ WNL.0b013e318225ab2e
- McDonald, C. M., Campbell, C., Torricelli, R. E., Finkel, R. S., Flanigan, K. M., Goemans, N., et al. (2017). Ataluren in Patients with Nonsense Mutation Duchenne Muscular Dystrophy (ACT DMD): a Multicentre, Randomised, Double-Blind, Placebo-Controlled, Phase 3 Trial. *The Lancet* 390 (10101), 1489–1498. doi:10.1016/S0140-6736(17)31611-2
- Mcdonald, C. M., Henricson, E. K., Abresch, R. T., Florence, J., Eagle, M., Gappmaier, E., et al. (2013). The 6-minute Walk Test and Other Clinical Endpoints in Duchenne Muscular Dystrophy: Reliability, Concurrent Validity, and Minimal Clinically Important Differences from a Multicenter Study. *Muscle Nerve* 48 (3), 357–368. doi:10.1002/ MUS.23905

- Mendell, J. R., Goemans, N., Lowes, L. P., Alfano, L. N., Berry, K., Shao, J., et al. (2016). Longitudinal Effect of Eteplirsen versus Historical Control on Ambulation in Duchenne Muscular Dystrophy. Ann. Neurol. 79 (2), 257–271. doi:10.1002/ana.24555
- Mendell, J. R., Rodino-Klapac, L. R., Sahenk, Z., Roush, K., Bird, L., Lowes, L. P., et al. (2013). Eteplirsen for the Treatment of Duchenne Muscular Dystrophy. Ann. Neurol. 74 (5), 637–647. doi:10.1002/ana.23982
- Mercuri, E., Bertini, E., and Iannaccone, S. T. (2012). Childhood Spinal Muscular Atrophy: Controversies and Challenges. *Lancet Neurol.* 11 (5), 443–452. doi:10.1016/S1474-4422(12)70061-3
- Mercuri, E., Bertini, E., Messina, S., Solari, A., D'Amico, A., Angelozzi, C., et al. (2007). Randomized, Double-Blind, Placebo-Controlled Trial of Phenylbutyrate in Spinal Muscular Atrophy. *Neurology* 68 (1), 51–55. doi:10.1212/01.wnl.0000249142.82285.d6
- Mercuri, E., Darras, B. T., Chiriboga, C. A., Day, J. W., Campbell, C., Connolly, A. M., et al. (2018). Nusinersen versus Sham Control in Later-Onset Spinal Muscular Atrophy. N. Engl. J. Med. 378 (7), 625-635. doi:10.1056/NEJMoa1710504
- Mercuri, E., Finkel, R., Montes, J., Mazzone, E. S., Sormani, M. P., Main, M., et al. (2016). Patterns of Disease Progression in Type 2 and 3 SMA: Implications for Clinical Trials. *Neuromuscul. Disord.* 26 (2), 126–131. doi:10.1016/j.nmd.2015.10.006
- Mohamed, K., Appleton, R., and Nicolaides, P. (2000). Delayed Diagnosis of Duchenne Muscular Dystrophy. *Eur. J. Paediatric Neurol.* 4 (5), 219–223. doi:10.1053/ejpn.2000.0309
- Muntoni, F., Domingos, J., Manzur, A. Y., Mayhew, A., Guglieri, M., Sajeev, G., et al. (2019). Categorising Trajectories and Individual Item Changes of the North Star Ambulatory Assessment in Patients with Duchenne Muscular Dystrophy. *PLoS ONE* 14 (9), e0221097–15. doi:10.1371/ journal.pone.0221097
- Muntoni, F., Signorovitch, J., Sajeev, G., Done, N., Yao, Z., Goemans, N., et al. (2021). EP.125 Minimal Detectable Changes in Functional Measures in Duchenne Muscular Dystrophy (DMD): A Study of Multiple Centers, Networks and Trial Arms. *Neuromuscul Disord*. 31, S87, 2021. Available at: http://www.nmd-journal.com/article/S0960896621003370/fulltext (Accessed June 28, 2021).
- Niks, E. H., and Aartsma-Rus, A. (2017). Exon Skipping: a First in Class Strategy for Duchenne Muscular Dystrophy. *Expert Opin. Biol. TherapyTaylor Francis Ltd* 17 (2), 225–236. doi:10.1080/14712598.2017.1271872
- Passamano, L., Taglia, A., Palladino, A., Viggiano, E., D'Ambrosio, P., Scutifero, M., et al. (2012). Improvement of Survival in Duchenne Muscular Dystrophy: Retrospective Analysis of 835 Patients. Acta Myol 31 (OCTOBER), 121–125.
- Powell, P. A., Carlton, J., Rowen, D., Chandler, F., Guglieri, M., and Brazier, J. E. (2021). Development of a New Quality of Life Measure for Duchenne Muscular Dystrophy Using Mixed Methods. *Neurology* 96 (19), e2438–e2450. doi:10.1212/WNL.000000000011896
- Ricotti, V., Mandy, W. P. L., Scoto, M., Pane, M., Deconinck, N., Messina, S., et al. (2016). Neurodevelopmental, Emotional, and Behavioural Problems in Duchenne Muscular Dystrophy in Relation to Underlying Dystrophin Gene Mutations. *Dev. Med. Child. Neurol.* 58 (1), 77–84. doi:10.1111/ dmcn.12922
- Ricotti, V., Ridout, D. A., Pane, M., Main, M., Mayhew, A., Mercuri, E., et al. (2016). The NorthStar Ambulatory Assessment in Duchenne Muscular Dystrophy: Considerations for the Design of Clinical Trials. J. Neurol. Neurosurg. Psychiatry 87 (2), jnnp-2014. doi:10.1136/jnnp-2014-309405
- Ricotti, V., Ridout, D. A., Scott, E., Quinlivan, R., Robb, S. A., Manzur, A. Y., et al. (2013). On Behalf of the NorthStar Clinical NetworkLong-Term Benefits and Adverse Effects of Intermittent versus Daily Glucocorticoids in Boys with Duchenne Muscular Dystrophy. J. Neurol. Neurosurg. Psychiatry 84 (6), 698–705. doi:10.1136/jnnp-2012-303902
- Ricotti, V., Selby, V., Ridout, D., Domingos, J., Decostre, V., Mayhew, A., et al. (2019).
 Respiratory and Upper Limb Function as Outcome Measures in Ambulant and Non-ambulant Subjects with Duchenne Muscular Dystrophy: A Prospective Multicentre Study. *Neuromuscul. Disord.* 29 (4), 261–268. doi:10.1016/j.nmd.2019.02.002
- Ryder, S., Leadley, R. M., Armstrong, N., Westwood, M., de Kock, S., Butt, T., et al. (2017). The burden, Epidemiology, Costs and Treatment for Duchenne

Muscular Dystrophy: An Evidence Review. Orphanet J. Rare Dis. 12 (1). doi:10.1186/s13023-017-0631-3

- Sarepta Therapeutics (2021). Sarepta Therapeutics Announces Top-Line Results for Part 1 of Study 102 Evaluating SRP-9001, its Investigational Gene Therapy for the Treatment of Duchenne Muscular Dystrophy. Available at: https://investorrelations.sarepta.com/news-releases/newsrelease-details/sarepta-therapeutics-announces-top-line-results-part-1study-102 (Accessed May 7, 2021).
- Servais, L., Straathof, C. S. M., Schara, U., Klein, A., Leinonen, M., Hasham, S., et al. (2020). Long-term Data with Idebenone on Respiratory Function Outcomes in Patients with Duchenne Muscular Dystrophy. *Neuromuscul. Disord.* 30 (1), 5–16. doi:10.1016/j.nmd.2019.10.008
- Shirley, M. (2021). Casimersen: First Approval. Drugs 81 (7), 875–879. doi:10.1007/ s40265-021-01512-2
- Spurney, C. F. (2011). Cardiomyopathy of Duchenne Muscular Dystrophy: Current Understanding and Future Directions. *Muscle Nerve* 44 (1), 8–19. doi:10.1002/ mus.22097
- Stolte, B., Bois, J. M., Bolz, S., Kizina, K., Totzeck, A., Schlag, M., et al. (2020). Minimal Clinically Important Differences in Functional Motor Scores in Adults with Spinal Muscular Atrophy. *Eur. J. Neurol.* 27 (12), 2586–2594. doi:10.1111/ ene.14472
- Sussman, M. D., Sienko, S. E., Buckon, C. E., Hilton, C., de Mattos, C. B., and d'Amato, C. (2020). Efficacy of Corticosteroid in Decreasing Scoliosis and Extending Time to Loss of Ambulation in a Single Clinic: An Effectiveness Trial. J. Child. Orthop. 14 (5), 421–432. doi:10.1302/ 1863-2548.14.200156
- Swoboda, K. J., Scott, C. B., Crawford, T. O., Simard, L. R., Reyna, S. P., Krosschell, K. J., et al. (2010). SMA CARNI-VAL Trial Part I: Double-Blind, Randomized, Placebo-Controlled Trial of L-Carnitine and Valproic Acid in Spinal Muscular Atrophy. *PLoS ONE* 5 (8), e12140. doi:10.1371/journal.pone.0012140
- Syed, Y. Y. (2016). Eteplirsen: First Global Approval. Drugs 76 (17), 1699–1704. doi:10.1007/s40265-016-0657-1
- Tuffery-Giraud, S., Béroud, C., Leturcq, F., Yaou, R. B., Hamroun, D., Michel-Calemard, L., et al. (2009). Genotype-phenotype Analysis in 2,405 Patients with a Dystrophinopathy Using the UMD-DMD Database: A Model of Nationwide Knowledgebase. *Hum. Mutat.* 30 (6), 934–945. :. doi:10.1002/ humu.20976
- Us Food and Drug Administration (2016). FDA grants Accelerated Approval to First Drug for Duchenne Muscular Dystrophy [Online]. Available at: https:// www.fda.gov/news-events/press-announcements/fda-grants-acceleratedapproval-first-drug-duchenne-muscular-dystrophy (Accessed May 28, 2021).
- U.S. Department of Health and Human Services and Food and Drug Administration (2019). Submitting Documents Using Real-World Data and Real-World Evidence to FDA for Drugs and Biologics Guidance for Industry DRAFT GUIDANCE. Available at: https://www.fda.gov/Drugs/GuidanceComplianceRegulatoryInformation/ Guidances/default.htm (Accessed July 19, 2021).
- U.S. Food and Drug Administration (2020). FDA Approves Targeted Treatment for Rare Duchenne Muscular Dystrophy Mutation [Online]. Available at: https://www. fda.gov/news-events/press-announcements/fda-approves-targeted-treatment-rareduchenne-muscular-dystrophy-mutation (Accessed July 1, 2021).
- van Ruiten, H. J. A., Straub, V., Bushby, K., and Guglieri, M. (2014). Improving Recognition of Duchenne Muscular Dystrophy: a Retrospective Case Note Review. *Arch. Dis. Child.* 99 (12), 1074–1077. doi:10.1136/archdischild-2014-306366
- Wadman, R. I., van der Pol, W. L., Bosboom, W. M., Asselman, F.-L., van den Berg, L. H., Iannaccone, S. T., et al. (2020). Drug Treatment for Spinal Muscular Atrophy Types II and III. Cochrane 1. in *Database Of Systematic Reviews [Online]* (John Wiley & Sons), 2020. doi:10.1002/ 14651858.CD006282.pub5
- Wadman, R. I., Wijngaarde, C. A., Stam, M., Bartels, B., Otto, L. A. M., Lemmink, H. H., et al. (2018). Muscle Strength and Motor Function throughout Life in a Cross-sectional Cohort of 180 Patients with Spinal Muscular Atrophy Types 1c-4. *Eur. J. Neurol.* 25 (3), 512–518. doi:10.1111/ene.13534
- Wagner, K. R., Kuntz, N. L., Koenig, E., East, L., Upadhyay, S., Han, B., et al. (2021). Safety, Tolerability, and Pharmacokinetics of Casimersen in Patients with D

Uchenne Muscular Dystrophy Amenable to Exon 45 Skipping: A Randomized, Double-blind, Placebo-controlled, Dose-titration Trial. *Muscle & Nerve* 64, 285–292. doi:10.1002/mus.27347

- Wave Life Sciences (2019). Discontinuation Of Suvodirsen Development For Duchenne Muscular Dystrophy. Available at: https://ir.wavelifesciences.com/ news-releases/news-release-details/wave-life-sciences-announces-discontinuationsuvodirsen (Accessed July 28, 2021).
- Yamamoto, T., Awano, H., Zhang, Z., Sakuma, M., Kitaaki, S., Matsumoto, M., et al. (2018). Cardiac Dysfunction in Duchenne Muscular Dystrophy Is Less Frequent in Patients with Mutations in the Dystrophin Dp116 Coding Region Than in Other Regions. *Circ. Genomic Precision Med.* 11 (1), e001782. doi:10.1161/CIRCGEN.117.001782
- Zhang, S., Qin, D., Qin, D., Wu, L., Li, M., Song, L., et al. (2021). Genotype Characterization and Delayed Loss of Ambulation by Glucocorticoids in a Large Cohort of Patients with Duchenne Muscular Dystrophy. Orphanet J. Rare Dis. 16 (1). doi:10.1186/s13023-021-01837-x

Conflict of Interest: GB has received consultancy honoraria from AveXis, Roche, Biogen, PTC, and Sarepta Therapeutics. GB has received speaker honoraria from AveXis, Roche and PTC. FM is supported by the NIHR Great Ormond Street Hospital Biomedical Research Centre and has received speaker and consultancy honoraria from Sarepta Therapeutics, Avexis, PTC Therapeutics, Roche and Pfizer.

The remaining authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's Note: All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Copyright © 2021 Stimpson, Chesshyre, Baranello and Muntoni. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.